# International Journal of Technology Assessment in Health Care

cambridge.org/thc

# Commentary

As members of the HTAi Patient and Citizen Involvement in HTA Interest Group, Patient Involvement and Education Working Group, Health Technology Assessment international (HTAi), 1200 10405 Jasper Avenue, Edmonton, Alberta, T5J 3N4, Canada.

**Cite this article:** Wale JL, Sullivan M (2020). Exploration of the visibility of patient input in final recommendation documentation for three health technology assessment bodies. *International Journal of Technology Assessment in Health Care* **36**, 197–203. https://doi.org/ 10.1017/S0266462320000240

Received: 13 March 2019 Revised: 15 March 2020 Accepted: 23 March 2020 First published online: 7 May 2020

#### Key words:

Health technology assessment; Patient involvement; Experiential knowledge; Patient group submissions; HTA appraisal

#### Abbreviations:

CADTH: Canadian Agency for Drugs and Technologies in Health; CADTH-CDEC: CADTH Canadian Drug Expert Committee; CADTH-CDR: CADTH Common Drug Review; CADTHpCODR: CADTH pan-Canadian Oncology Drug Review; CADTH-pERC: CADTH pCODR Expert Review Committee; CCC: Crohn's and Colitis UK; HTA: Health Technology Assessment; HTAi: Health Technology Assessment International; IBD: Inflammatory bowel disease; NICE: National Institute for Health and Care Excellence; OS: Overall survival; PACE: Patient and Clinician Engagement (PACE)—Scottish Medicines Consortium meeting to gather input for orphan/ultra-orphan and end of life medicines: PCIG: Patient and Citizen Involvement Interest Group of HTAi; QoL: Quality of life; SMC: Scottish Medicines Consortium

© Cambridge University Press 2020. This is an Open Access article, distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivatives licence (http:// creativecommons.org/licenses/by-nc-nd/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is unaltered and is properly cited. The written permission of Cambridge University Press must be obtained for commercial re-use or in order to create a derivative work.



# Exploration of the visibility of patient input in final recommendation documentation for three health technology assessment bodies

# Janet L. Wale 💿 and Melissa Sullivan<sup>2</sup>

HTAi Patient and Citizen Involvement in HTA Interest Group, 11A Lydia Street, Brunswick, Victoria 3056, Australia and <sup>2</sup>Patient Engagement in HTA, HTAi Patient Involvement and Education Working Group, Coordinator, 50 Portugal Cove Road, St. John's, NL A1B 2L9, Canada

Health technology assessment (HTA) recommendations informed by patient concerns are seen to ensure democracy and legitimacy. We explored how written and oral patient involvement in two HTAs was reported on in publicly available final recommendations and discussion summaries of appraisal committees from three HTA bodies. We aimed to gain insights into how patient input was utilized by appraisal committees to better understand the goals of patient involvement and how these are being achieved. In each of the three HTA bodies, templated submission questionnaires provide a formal process for seeking written patient group input. Additionally, the National Institute for Health and Care Excellence (NICE) selects patient experts to provide a templated submission and attend appraisal committee meetings. For Scottish Medicines Consortium (SMC), a patient advocate and clinician combined meeting (PACE) discussed the cancer drug, referred to in the final recommendation. The discussion summaries of all appraisal committees contained references to patient involvement. Where two mechanisms for patient involvement were provided, oral input from the expert patients and PACE were more clearly documented than information from written patient group submissions. NICE reports focused on the perspective of the patient expert. The SMC report highlighted feedback from the PACE throughout. We suggest that the lack of clear reporting on the use of patient group input in deliberations and therefore accountability to patient groups limits progress in patient involvement in HTA. Patient groups may therefore not have a clear understanding of what information they can best provide to inform deliberations, and in reporting back to members.

Health technology assessment (HTA) requires rigorous analytical methods with an emphasis on randomized controlled trials and appraisal of the evidence for benefits, harms, and costs by expert committees, but a focus on clinical outcomes alone may not be sufficient (1). Important information relevant to the clinical use of technology includes patient-relevant health outcomes, access to healthcare services, patient preferences, quality of life, immediate and longer-term consequences (2;3). This is in addition to burdens and costs for users of the technologies and overall patient and carer experiences. Patients can provide unique, experiential knowledge to give context to the clinical research, help to overcome uncertainties in clinical evidence, and inform the determination of the value of a technology or service (4–7).

Patients' views may differ from those of health professionals or researchers and cannot be assumed by other stakeholders (2;8;9). Patient involvement has a key role in the legitimacy of HTAs such that it is important that HTA recommendations have been informed by patient concerns and perspectives (10). The rights and value of patients participating in HTAs are becoming more widely supported (11). It is therefore important, for public understanding and knowledge, that the involvement process supports learning through informed conversations (12) and transparent reporting on HTAs (13).

Quality criteria have recently been published for patient engagement in drug development, calling for shared purpose and respect for the resources and energy required for providing patient input. The criteria include representativeness of stakeholders with agreed roles and responsibilities, the resources and capabilities to enable meaningful engagement, transparency in communication and documentation, and continuity and sustainability (13). The opportunity for learning from feedback is important for patient advocates and patient groups involved in HTA. Indeed, it is likely that many patient groups do not have a complete understanding of what information they might need to convey, what would be of most use in HTA deliberations, the form that needs to take, and which insights if any could impact the deliberations (10). The need for learning experiences is often not acknowledged by those who are in a position to provide the feedback (14).

Boothe (10) identified that goals for patient involvement in HTA continue to be unclear, without a clear vision of what the involvement should achieve. A lack of alignment on thinking

among the different players also exists, making it very difficult to meaningfully evaluate patient involvement. The author used surveys and interviews over a number of years to describe how the different players in HTA appraisals saw the goals of patient group involvement, and how thinking had changed over time, for the Ontario Committee to Evaluate Drugs and the pan-Canadian Common Drug Review (CADTH-CDR), now part of the Canadian Agency for Drugs and Technologies in Health (CADTH) (10).

The HTAi Patient and Citizen Involvement in HTA Interest Group (PCIG) previously collated the experiences of international patient advocates in providing input to the appraisals of health technologies for HTA bodies. We identified a common lack of clearly-defined purpose and goals of patient input, limited feedback to advocates on the value and use of the patient input, and limited communication to the public of the impact of patient involvement (15). Only the patient advocates who were patient representatives on an HTA appraisal committee were able to refer to specific positive recommendations where important contributions were made through patient group submissions, for example, on quality of life and more adversely affected sub-populations (15). In recent years, we have identified patient advocate concerns about how patient involvement in HTA appraisals is reported on, leading to uncertainties about how the input was used. In this commentary, we explore this concern by looking at how three HTA bodies reported on their use of patient involvement in final appraisal committee recommendations and discussion summaries. Our aim was to elucidate how the HTA bodies described their use of patient input in the final recommendations and discussion summaries, and so to observe the situation and view it as unique without imposing any deductive structure upon it. Such insights into how patient input was utilized may help us better understand the goals of patient involvement and how these are being achieved by HTA bodies. We see the need to examine real-world evidence on the value of patient input as is currently being used. Without clear reporting on the use of patient group input patient groups may not have a clear understanding of what information they can best provide to inform deliberations. Providing this evidence may also lead to more open, meaningful discussions and work on the culture and practice around the utilization and reporting of patient input, to encourage learning and evaluation of methods used to gain patient input into HTAs.

We worked with three HTA bodies that have formal processes to collect patient group input through submission templates with carefully defined questions. An important difference among the processes used is that the National Institute for Health and Care Excellence (NICE) and Scottish Medicines Consortium (SMC) ask for the input during the appraisal stage whereas CADTH asks for the input during protocol development. NICE also has a process for involving patient groups in discussions to inform the development of the protocol. At appraisal, NICE invites patient experts generally nominated by patient groups to report on their experiences with the disease and its treatment, and to answer questions. SMC provides the opportunity for technology sponsors to request a Patient and Clinician Engagement meeting (PACE) before its appraisal of orphan or ultra-orphan and end of life medicines; a summary of the meeting is presented at the SMC committee meeting alongside the patient group submissions. We selected two different pharmaceutical interventions for two different medical conditions, as a single example could more easily be considered as an anomaly or specific to a particular health condition.

# **Methods for Information Gathering**

For each of the three HTA bodies, a final committee discussion report summarizes the deliberations of the appraisal committee that lead to the final recommendation (Figure 1). By "final recommendation" we refer to the funding recommendation (CADTH), guidance (NICE), or advice (SMC) from the appraisal committees, together with the summary committee discussion report. SMC incorporates both into a "Detailed Advice Document"; and CADTH Canadian Drug Expert Committee (CADTH-CDEC) into "Recommendations and Reasons" (see Figure 1). As an exploratory study, we descriptively analyzed the documentation for explicit and inferred reference to the different types of patient input, particularly in describing the needs and concerns of patients and caregivers and what they value. Inferred contributions related particularly to the descriptions of the health condition, impact of the disease, and its treatment. Extracts of the text were entered manually into Excel by the first author and checked by the second author. Although the submission templates provide for input under specific headings, no attempt was made to go beyond these descriptors with a thematic qualitative analysis or to give weight to the input beyond what was recorded. We sought feedback from the other members of the HTAi PCIG working group when summarizing this process and their comments were diligently addressed.

We considered two drug appraisals by each of the three HTA bodies (16–21), for treatment of a chronic disease (ustekinumab) and cancer (blinatumomab). Appraisals took place from May 2016 to August 2017. We summarized, for comparison with the publicly available appraisal committee discussion reports, the key points made by patient groups in their templated submissions: on how the condition affects patients' daily lives; experience with symptoms and severity of the condition; how patients are affected by current treatments, and how well they are managing; impact on caregivers; experience with or expectations of a new treatment in terms of benefits, advantages, and disadvantages compared to current treatments; impact on family and caregivers; whether the new treatment is innovative; and any other factors.

# HTA Appraisal of a Chronic Disease Treatment

Ustekinumab is a drug for moderately to severely active Crohn's disease in adults whose previous treatment resulted in an inadequate response, a response that was not maintained, or where people could not tolerate or have contraindications to the treatment. Previous treatments were conventional therapy or a TNF- $\alpha$  inhibitor biologic.

### Patient Group Templated Submission

The patient groups that provided patient input were: CADTH, Crohn's and Colitis Canada (CCC) and GI (Gastrointestinal) Society, Canada; NICE, Crohn's and Colitis UK; and SMC, Crohn's and Colitis UK. The patient groups stated their sources of information and the submissions were rich with patient quotes (Figure 2). The NICE and SMC submissions were provided by the same UK organization but were substantially different from each other.

#### Final Recommendations

For each HTA body, the final recommendations did not directly reference the patient input received, and the cost was a significant factor in the recommendations. We then looked at the reports of the appraisal committee deliberations for reference to patient input.

CADTH-CDEC
Recommendation and Reasons
Patient summary in 'Summary of CDEC Considerations' section
Committee papers to inform deliberations - clinical and economic reports, linked to on webpage
Patient group submission, linked to on webpage
CADTH-pERC
Final recommendation
Summary of pERC Deliberations, 'Evidence in Brief' includes: Patient-based values – a summary of patient group input
Final Clinical Guidance Report – to inform committee deliberations, linked to on webpage and has the patient group submissions
Patient Group Conflict of Interest Declarations, linked to on webpage
NICE
Recommendations
Evidence – with links to committee papers
<ul> <li>Patient main points in box labelled 'the patient's perspective'</li> </ul>
Patient group submission (Appendix G) and patient expert personal statements (Appendix D) are in the committee papers prepared for the committee deliberations
SMC
The 'Detailed Advice Document' is publically available and contains a
- summary of patient and public involvement, with a high level of detail.
PACE report included in papers to inform committee deliberations
Patient group submissions are not on the website

**Figure 1.** Website presentation of the final recommendations and committee discussion reports, and patient-relevant submissions, for each HTA body: Canadian Agency for Drugs and Technologies in Health—Canadian Drug Expert Committee (CADTH-CDEC), Canadian Agency for Drugs and Technologies in Health—pan-Canadian Oncology Drug Review Expert Review Committee (CADTH-pERC), National Institute for Health and Care Excellence (NICE) and Scottish Medicines Consortium (SMC).

## Final Committee Discussion Reports

For CADTH, the "Recommendations and Reasons" stated that the "Canadian Drug Expert Committee (CDEC) considered patient group submitted information prepared by the Common Drug Review (CDR) on outcomes and issues important to individuals living with Crohn's disease." The clinical and economic reports provided to CDEC were outside this study remit (Figure 1) and we did not look at any use of the patient group submissions to interpret the clinical trial results and critique the economic model provided to the appraisal committee.

A summary of the patient group submission to NICE was given. Two patient experts presented to the NICE appraisal committee. The committee discussion report focused on the patient experts: "It heard from the patient experts that..."; "The patient experts recounted their experience of"; "both the patient experts had ustekinumab as part of a clinical trial after losing response to TNF-alpha inhibitors, being able to '...resume work and everyday activities as well as avoiding the need for surgery'." One patient expert explained that ustekinumab had left him feeling "wonderfully normal again" while the other highlighted the immense improvement to her quality of life, enabling her to start a family. The patient experts emphasized that maintenance treatment with ustekinumab is a subcutaneous injection rather than an intravenous infusion, which is greatly valued by patients because it means they can take the treatment at home with no need for hospital visits. The key conclusions state: "The committee acknowledged that ustekinumab is a convenient and well-tolerated treatment that has considerably improved the quality of life of the patient experts."

For SMC, the patient group templated submission was summarized as a bulleted list in a "Summary of patient and public involvement" section of the final appraisal committee discussion report.

#### Inferred Use of Patient Input

Looking at inferred use of patient group submissions, examples were: "The [NICE] committee understood that Crohn's disease follows an unpredictable, relapsing and remitting course with many debilitating symptoms"; "The [NICE] committee heard that patients fear loss of remission and exacerbations of the disease because of the major impact these have on quality of life... it is very important to have a range of treatment options ...."

# HTA Appraisal of a Cancer Treatment

Blinatumomab was the drug under consideration for adults who had relapsed or were refractory to previous chemotherapy treatment.

### Patient Group Templated Submission

The patient groups that provided patient input were: CADTHpCODR, Leukemia & Lymphoma Society of Canada; NICE, Leukaemia Care; and SMC, Leukaemia Care.

Direct quotes nom patient group templated submission	Direct qu	otes from	patient gr	oup templa	ated submissions
--	-----------	-----------	------------	------------	------------------

"Having Crohn's that is in control is fairly routine, it doesn't affect my life. However, flaring Crohn's is another story. I experience debilitating pain every day, I am constantly exhausted, I am careful of what I eat, cannot eat/drink what I like and I always have to know where the toilet is. I have had numerous accidents where I have not been able to make it to the bathroom on time, this is embarrassing. I am hopeful that better treatment will come for us suffering from this condition."

"I am 23 yo and I have had to leave my university place 3 times due to my Crohn's disease. My life has been on hold for years due to this illness and I have lost 3y of income, which has been a great burden."

(Crohn's and Colitis UK)

"Every day was a struggle. I went to the washroom 23 times a day or more and got up every hour of the night with night sweats and severe cramping. My Crohn's made me so ill that I required numerous surgeries. Once my surgeon even said I was lucky to end up in ED as wi/i another week I might not have survived.."

"My energy levels have decreased and I get fatigued much more easily, the fear of pain, bleeding, incontinence is horrible. The worst part is fearing the next big flare that will prevent me from being a mom to my 18m old.."

"..surgery is the last option that I would want to go through. Each time I go through surgery it gets harder because they are removing more and more bowel. You get faced with other challenges and more problems each time because now your bowels are shorter and shorter."

(GI Society Canada)

Figure 2. Direct patient quotes from patient group templated submissions for moderately to severely active Crohn's disease.

#### Direct quotes from patient group templated submissions

"..he has lost his interest in hobbies, his will to enjoy life, his will to travel, his interest to do anything else besides lying on the sofa"

"Blincyto has been the only positive of all the treatments so far"; "..has improved my quality of life compared to previous therapies that I used"

Leukemia & Lymphoma Society of Canada

Figure 3. Direct quotes from patient group templated submissions for previously treated Philadelphia-chromosome-negative acute lymphoblastic leukemia and blinatumomab.

The patient groups stated their sources of information. Three patients had direct experience of the new treatment in Canada. In England and Scotland, the patient group was unable to identify any patients with direct experience. The patient group submissions contained few quotes (Figure 3). The NICE and SMC submissions were from the same patient group and their content was similar. For NICE and SMC, the submissions emphasized the importance to patients of being able to have a treatment-free period and the option for some patients to be treated as outpatients, allowing patients to spend more time at home with their families. For CADTH, the patient group submission concentrated

on troublesome symptoms and side effects for six people who had not taken the drug and on patient experience for the three who had received blinatumomab.

# Final Recommendations

The CADTH pan-Canadian Oncology Drug Review (CADTHpERC) and NICE final recommendations did not refer to specific patient input whilst the SMC Advice statement clearly referred to PACE: "This advice takes account of the views from a Patient and Clinician Engagement (PACE) meeting."

#### Final Committee Discussion Reports

For CADTH, the "Summary of pERC Deliberations" states that "The Committee deliberated on the patient advocacy group input, which indicated that patients with ALL value disease control and the management of side effects related to current therapies for ALL. Given improvements in OS and maintenance of QoL, pERC agreed that blinatumomab aligns with patient values overall."

For NICE, it is the perspective of the patient expert that is given with details about the physical and psychological stresses of failing chemotherapy, and the emotional strain on families and friends: "The committee heard from a patient expert that.."; "The committee heard from the patient expert that patients whose disease responds to blinatumomab can live a relatively normal life during treatment, with minimal side effects." In some of the text, patient expert and clinical expert views were linked: "...heard from the patient expert and clinical experts that people with ....."

The committee concluded in their deliberations that the availability of an effective treatment that could be delivered primarily in the outpatient setting was hugely beneficial to patients and would have a major impact on their quality of life. A summary of the patient group submission was provided.

For SMC, the patient group submission was referred to and a detailed summary of the submission provided. It was stated that a patient group submission was received from Leukaemia Care, who provided representatives to attend PACE, and that points of the submission were covered under PACE. Detailed information was given on PACE deliberations in relation to "the relatively young age of patients (mid to late thirties) and the high symptom burden associated with this condition and how a diagnosis of relapse is devastating for patients and their families." At the PACE meeting, "the deep response and high remission rates with blinatumomab, a greater chance of moving to allogeneic haematopoietic stem cell transplantation, improved quality of life and more time spent at home with the family" were noted. It is evident that the PACE feedback was very seriously considered in discussions and was integrated throughout the report.

## Inferred Use of Patient Input

Patient input was inferred in the following examples from NICE: "The committee acknowledged that acute lymphoblastic leukaemia does not affect the patient in isolation, but also places emotional strain on their families and friends"; "Living with precursor B-cell acute lymphoblastic leukaemia, particularly when the condition has failed to respond to first line chemotherapy, can have a profound effect on a person's physical and psychological wellbeing."

#### Discussion

Our exploration of the final recommendations and summary committee discussion reports for the appraisal of two drugs across three HTA bodies showed clear differences in how the types of patient input—patient group submissions, patient experts, and patient and clinician meetings—were referred to. One possible interpretation is that each type of patient input was considered differently by the appraisal committees. Differences could be caused, for example, by variations in how the different types of input were presented, who presented it, and at what stage of the meeting, and if and how the input resonated with expert members of the appraisal committees. Because no direct feedback was given by HTA agencies to patient groups, it would be very difficult for the patient groups to know how their input was received and how useful it was; which may align with a lack of clear expectations for patient submissions. On the other hand, it could be that report writers are not being asked to specifically capture the utilization of all types of patient input in the deliberations, and a different set of skills may be needed to do this, whilst maintaining the objectivity of the HTA process.

Overall, direct patient input through participation in meetings appeared to be more influential than written patient group submissions. Patient input was utilized by CADTH-pERC to inform alignment with patient values, with little further information on how "alignment with patient values" actually influenced the deliberative process. For NICE, the perspectives of patient experts were referred to in illustrating discussion points, with limited reference to the patient group submissions. For SMC, the PACE feedback was referred to and integrated throughout the summary discussion report. The collective views of patient advocates and clinicians and how they were presented to the appraisal committee would appear to have been highly valued by the committee. Indeed, patient involvement may work best when health professionals and patients enter into dialogue to share their respective expertise and learn from each other (12).

If experiential knowledge gained from the patient group submissions was being utilized in deliberations, this use may not always be clearly recorded. We did not observe any attempt to relate one form of patient involvement to another form; and any use of experiential patient knowledge to interpret the clinical trial findings was not evident in the reports, including for quality of life data. It has been suggested that better engagement of patients, users, caregivers, citizens throughout the assessment and appraisal process to seek convergence of scientific data on efficiency, safety, and acceptability with contextual and experiential data could be used to strengthen decision making (7). For NICE, rich information from patient experts could help overcome the limitations of clinical trials where patient-reported outcomes, patient experience, or quality of life data were missing or the measures were too generic (22). A study of patient engagement in CADTH-pCODR found that submissions from patient advocacy groups were most impactful when they provided information that was not available, well documented, or easily recognized from other sources. This included information that was not collected in clinical trials, any clinical trade-offs, information on lived experience, and patient and caregiver priorities. The committee gained a better understanding of the realities of a disease and its practical burdens from patient input (6).

Staley and Barron (12) have identified the importance of mutual learning as an outcome of patient involvement. Quality interactions can lead to greater understanding and better decisions through respectful conversations with a two-way exchange of knowledge. Patient experts are present during discussions but they may not be charged with the ability, confidence, and opportunity to challenge committee members' assumptions and perspectives in sharing their experiential knowledge. Recording where patient input has an impact supports wider learning as it can fill gaps in the knowledge of other stakeholders and correct or modify assumptions (12). The overall uncertainty about the specific roles of the different types of patient involvement and how they work together is likely to result in difficulties in strengthening the value of patient involvement, and in informing changes in thinking (10) as we move toward more personcentered health care. Completing patient group submissions is dependent on the goodwill of patients. If the use of these submissions is not well documented, the ethics of HTA bodies asking for such submissions is questionable (23).

201

In-person attendance at appraisal committee meetings as observers (22) and from patient advocates on appraisal committees (15) has clearly shown that patient input is discussed and that patient experiences and values are important. CADTH researchers were able to track insights contained in patient group submissions to the Common Drug Review before the assessment reports were developed through to the appraisal process and recommendations. They concluded that the points made were included in framing the assessment and in interpreting the evidence, and so were integrated throughout an HTA (24). On the other hand, in a study of patient experts' perspectives, the authors found that the committee processes could undermine the input from the patient experts, leading them to have a largely peripheral consultative role (22). The latter authors suggested that a change in culture of the appraisal committees may be needed, and that having patients present to provide their insights increased the level of complexity that committee members faced. Clinicians and HTA professionals can play a major role in excluding, reshaping, or admitting the more subjective experientially based input from patients and carers (22). An environment is needed where a diversity of views can be readily articulated for patient involvement to meet the objectives of an organization (25). Boothe (10) was able to show that more dialogue is now possible with technical experts on the goals, strengths, and challenges with patient involvement, in Canada.

The HTA bodies involved in our study have patient and public involvement teams whose job it is to work with patient groups to support and coordinate patient input into HTAs. Since July 2017, SMC now invites a patient group representative from each submitting patient group to participate during the meeting, and is introducing training sessions for its committee members on working with patient advocates. SMC and NICE appraisal committee meetings are open to the public but with a closed session for decision making; and both produce a public summary document to explain the decisions of the appraisal committees. CADTH is looking at how feedback on the usefulness of the submissions can be given to patient groups. These actions and awareness of the present study may impact on the representation of patient input in the final committee discussion reports.

Limitations of our exploratory study therefore include the following. The appraisals that we studied were from May 2016 to August 2017, and it would be of value to investigate any changes since then. The HTA bodies involved in this study use similar online templates for collecting patient group submissions. These guide the content of the submissions. Our analysis of inferred reference to patient input could suggest that patient group submission content was frequently referred to in this way in the final documentation. This could be the result of how the report authors approached their task, resulting in the lack of visibility on where the content was from. NICE and SMC ask for patient submissions at the appraisal stage whereas CADTH asks for the input earlier to help inform researchers in developing the assessment protocols and throughout the assessment and appraisal processes. For CADTH-pERC, the patient input is incorporated into addressing "patient values." We analyzed the final committee discussion reports summarizing the deliberations of the appraisal committees. The papers provided to the appraisal committees for their meetings were not part of the analysis. Finally, our investigation is an exploration of only two case studies. The pharmaceuticals were appraised using standard processes and the quality of patient input was not a criterion for selecting the case studies.

Impactful patient involvement in HTAs requires the confidence of patient groups that their input is an important part of the deliberations of appraisal committees, and that the groups can be seen as partners in the appraisal of new technologies. This study identified limited referral to the utilization of patient group templated submissions in the final appraisal committee discussion reports, with greater use of information from patient experts (NICE) and PACE (SMC). Our exploration of how HTA bodies described their use of patient input in the final recommendations and discussion summaries of HTA appraisals highlights that it is important to be able to facilitate improvements in current approaches to the reporting of patient input in HTAs, and to be able to readily identify where and how patient input has provided added value or impact.

We have assumed, by developing templates for patient groups to complete and summarizing this information in appraisal committee papers, that we know the aspects of patient experience that are important for HTA researchers and appraisal committees. Direct comparisons of what knowledgeable patient groups consider important, how they present their information in templates, and what the appraisal committees discuss are lacking in the literature. Our insights suggest that the most meaningful and efficient ways of presenting patient input need further investigation, as do the different types of patient input and the limitations placed on each type. For example, patient experts (NICE) may be in a position to provide illustrative stories and their own words but they are present to answer questions from the appraisal committee, in often intimidating circumstances (22). Management of real and perceived conflicts of interest of all stakeholders also needs consideration. The work of PACE (SMC) may demonstrate the potential added benefits of medical practitioners and patient advocates working together collaboratively, supported by experienced technical HTA staff, and is worthy of further investigation. A danger exists of added inefficiencies in regulation and reimbursement if the input of patient knowledge is restricted, including into the symptoms of a disease and presentation of clinical trial data on clinical outcomes. This is particularly so in lesser clinically known health conditions (26).

Acknowledgments. We thank the members of the HTAi Patient and Citizen Involvement in HTA Interest Group Patient Involvement and Education Working Group for their invaluable contributions to this project. We thank the patient organizations that gave SMC permission to share their submissions with the authors (Crohn's and Colitis UK and Leukaemia CARE). Submissions are presented at SMC meetings at a time when the meetings are open to the public. The same patient organizations provided submissions to NICE and these are publicly available on the NICE webpages. This work was presented as part of a panel session at the HTAi 2018 Annual Meeting. The authors thank our peer reviewers for their very helpful comments.

**Financial Support.** This research was not supported by any specific grant from any funding agency, commercial or not-for-profit sector organization.

**Ethics.** This study did not directly involve patients and carers. Patient group submissions were available on the websites of CADTH and NICE. SMC sought permission from the patient groups before sharing their submissions with the authors. These groups were Crohn's and Colitis UK and Leukaemia CARE; which also provided submissions to NICE. The patient input was provided by registered patient groups and involved people selected to present the views of patients in their capacity as patient advocates.

#### References

- Drummond MF, Schwartz JS, Jonsson B, Luce BR, Neumann PJ, Siebert U, et al. Key principles for the improved conduct of health technology assessments for resource allocation decisions. *Int J Technol Assess Health Care*. 2008;24:244–58.
- 2. Gagnon MP, Desmartis M, Lepage-Savary D, Gagnon J, St-Pierre M, Rhainds M, et al. Introducing patients' and the public's perspectives to

health technology assessment: A systematic review of international experiences. Int J Technol Assess Health Care. 2011;27:31–42.

- Mamzer M-F, Dubois S, Saout C; participants of Round Table of Giens XXXIII, et al. How to strengthen the presence of patients in health technology assessment conducted by the health authorities. Therapie. 2018;73:95–105.
- Facey K, Boivin A, Gracia J, Hansen HP, Lo Scalzo A, Mossman J, et al. Patients' perspectives in health technology assessment: A route to robust evidence and fair deliberation. *Int J Technol Assess Health Care*. 2010;26:334–40.
- Hofmann B, Cleemput I, Bond K, Krones T, Droste S, Sacchini D, et al. Revealing and acknowledging value judgments in health technology assessment. Int J Technol Assess Health Care. 2014;3:579–86.
- Rozmovits L, Mai H, Chambers A, Chan K. What does meaningful look like? A qualitative study of patient engagement at the Pan-Canadian Oncology Drug Review: Perspectives of reviewers and payers 2018. J Health Serv Res Policy. 2018;23:72–79.
- Roy M, Ganache I, Dagenais P. Advocating for a better engagement of patients, users, caregivers, and citizens in healthcare and social services technology assessment (HSTA). Commentary on "Assess, triangulate, converge, and recommend (ATCR): A process for developing recommendations for practice in the social sector using scientific, contextual and experiential data". *Int J Hosp Based Health Technol Assess*. doi:10.21965/ IJHBHTA.2018.002
- Staley K, Doherty C. It's not evidence, it's insight: Bringing patients' perspectives into health technology appraisal. *Res Involv Engagem*. 2016;2:4.
- 9. Dipankui MT, Gagnon MP, Desmartis M, Legare F, Piron F, Gagnon J, et al. Evaluation of patient involvement in a health technology assessment. *Int J Technol Assess Health Care*. 2015;**31**:166–70.
- Boothe K. "Getting to the table": Changing ideas about public and patient involvement in Canadian drug assessment. J Health Polit Policy Law. 2019;44:631–63.
- 11. Wale J, Scott AM, Hofmann B, Garner S, Low E, Sansom L. Why patients should be involved in health technology assessment. *Int J Technol Assess Health Care*. 2017;33:1–4.
- Staley K, Barron D. Learning as an outcome of involvement in research: What are the implications for practice, reporting and evaluation? *Res Involv Engagem*. 2019;5:14.
- Deane K, Delbecque L, Gorbenko O, Hamoir AM, Anton Hoos A, Nafria B, et al. Co-creation of patient engagement quality guidance for medicines development: An international multistakeholder initiative. *BMJ Innov.* 2019;5(1):43–55.
- 14. Mathie E, Wythe H, Munday D, Millac P, Rhodes G, Roberts N, et al. Reciprocal relationships and the importance of feedback in patient and

public involvement: A mixed methods study. *Health Expect.* 2018;21: 899–908.

- Scott AM, Wale JL. Patient advocate perspectives on involvement in HTA: An international snapshot. *Res Involv Engagem*. 2017;3:2.
- Canadian Agency for Drugs and Technologies in Health (CADTH) Common Drug Review: ustekinumab. 2016. Available from: https://www. cadth.ca/ustekinumab-15 (accessed April 11, 2018).
- National Institute for Health and Care Excellence (NICE) Ustekinumab for moderately to severely active Crohn's disease after previous treatment. 2017. Available from: https://www.nice.org.uk/guidance/ta456 with the submission on. Available from: https://www.nice.org.uk/guidance/ta456/ evidence (accessed April 11, 2018).
- Scottish Medicines Consortium (SMC) Stelara (ustekinumab) for Crohn's Disease. 2017. Available from: http://www.scottishmedicines. org/files/advice/ustekinumab\_Stelara\_FINAL\_June\_2017\_for\_website.pdf (accessed April 11, 2018).
- Canadian Agency for Drugs and Technologies in Health (CADTH) pan-Canadian Oncology Drug Review Blincyto (blinatumumab) for acute lymphoblastic leukemia (Resubmission). 2017. Available from: https://www.cadth.ca/blincyto-acute-lymphoblastic-leukemia-resubmissiondetails (accessed April 11, 2018).
- 20. National Institute for Health and Care Excellence (NICE) Blinatumomab for previously treated Philadelphia-chromosome-negative acute lymphoblastic leukaemia. 2017. Available from: https://www.nice. org.uk/guidance/ta450 and the submission is on. Available from: https:// www.nice.org.uk/guidance/ta450/evidence (accessed April 11, 2018).
- Scottish Medicines Consortium (SMC) Blincyto (blinatumumab) for acute lymphoblastic leukaemia. 2016. Available from: http://www.scottish medicines.org/files/advice/DAD\_blinatumomab\_Blincyto\_FINAL\_May\_ 2016\_for\_website.pdf (accessed April 11, 2018).
- Hashem F, Calnan MW, Brown PR. Decision making in NICE single technological appraisals: How does NICE incorporate patient perspectives? *Health Expect.* 2018;21:128–37.
- Vanstone M, Abelson J, Bidonde J, Bond K, Burgess R, Canfield C, et al. Ethical challenges related to patient involvement in health technology assessment. Int J Technol Assess Health Care. 2019;35:253–56.
- Berglas S, Jutai L, MacKean G, Weeks L. Patients' perspectives can be integrated in health technology assessments: An exploratory analysis of CADTH Common Drug Review. *Res Involv Engagem*. 2016;2:21.
- Gibson A, Britten N, Lynch J. Theoretical directions for an emancipatory concept of patient and public involvement. *Health* (*London*). 2012;16:531–47.
- Falchetto R. The patient perspective: A matter of minutes. *Patient*. 2019;12:429–35.